\documentclass[11pt,twoside]{article}
%\usepackage{lgrind}    
%\usepackage{latexcad}   
\newcommand{\bs}{$\backslash$}
\newcommand{\theorem}{\bf Theorem}
\usepackage{amsfonts}
\usepackage{mathrsfs}
\usepackage{graphicx}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsthm} %
\usepackage{txfonts}
\usepackage{fontenc}
%\usepackage{stmaryrd}
\newtheorem{thm}{Theorem}[section]
%\theoremstyle{definition}
\newtheorem{defn}{Definition}[section]
\theoremstyle{remark} 
\newtheorem{note}{Note}[section]
\theoremstyle{plain}
\newtheorem{lem}[thm]{Lemma}
\newtheorem{obs}[thm]{Observation}
\newtheorem{cor}[thm]{Corollary}
\newtheorem{rmk}[thm]{Remark}
\newtheorem{prop}[thm]{Proposition}
\newtheorem{exam}[thm]{Example}
\renewcommand{\thethm}{\thesection.\arabic{thm}.}
%\renewcommand{\thelem}{\thesection.\arabic{lem}.}
%\renewcommand{\thecor}{\thesection\arabic{cor}.}
%\renewcommand{\thermk}{\thesection.\arabic{rmk}.}
%\renewcommand{\theprop}{\thesection.\arabic{prop}.
\pagestyle{myheadings}
\hbadness = 10000
\tolerance = 10000
\pretolerance = 10000
\renewcommand{\baselinestretch}{1.2}
\renewcommand{\arraystretch}{1}
\newcommand{\NI}{\noindent}
\voffset=-12mm
\mathsurround=2pt
\parindent=12pt
\parskip=6pt
\oddsidemargin=8mm
\evensidemargin=8mm
\topmargin=20pt
\headheight=12pt
%\footheight=12pt
\footskip=30pt
\textheight=230mm
\textwidth=150mm
\raggedbottom
\def\undertilde{\mbox{$\begin{array}{c}\vspace*{2pt}\\ { } ~\\
\hspace*{-4mm}  \sim \end{array}$}\hspace*{-8mm}}
\newcommand{\eb}{\vrule width 0.22 true cm height 0.22 true cm depth 0pt}
% \eb = Ende Beweis
\def \dis{\displaystyle}
\def \ra{\rightarrow}
\renewcommand{\thethm}{\thesection.\arabic{thm}}
%\renewcommand{\thelem}{\thesection.\arabic{lem}}
%\renewcommand{\thecor}{\thesection.\arabic{cor}}
%\renewcommand{\theprop}{\thesection.\arabic{prop}}
%\renewcommand{\theobs}{\thesection.\arabic{obs}}
%\renewcommand{\theequation}{ \thesection.\arabic{equation}}
%\pagestyle{myheadings}
\hbadness = 10000
\tolerance = 10000
\pretolerance = 10000
\usepackage[dvips]{color}
\setcounter{footnote}{1}
\usepackage{pstricks}
\begin{document}
\begin{thebibliography}{99}

\bibitem{lso1} Y. Alizadeh, M. Azari, T. Dosli\'c, Computing the eccentricity-related invariants of single-defect carbon nanocones, J. Comput. Theor. Nanosci., 10(6)(2013) 1297-1300.

\bibitem{h3} L. Barrier, F. Comellas, C. Dalfo and M.A. Fiol, The hierarchical product of graphs, Discrete Appl.  Math., 157(2009) 36-48.

\bibitem{h4} L. Barrier, C. Dalfo,  M.A Fiol and M. Mitjana,, The generalized  hierarchical product of graphs, Discrete Appl. Math., 309(2009) 3871-3881. 

\bibitem{i1} G. Chartrand, O.R. Oellermann, Applied and Algorithmic Graph Theory, McGraw-Hill, New York, 1993.

\bibitem{t3} P. Dankelmann, W. Goddard and C.S. Swart, The average eccentricity of a graph and its subgraphs, Util. Math., 65(2004) 41-51. 

\bibitem{lso8} H. Dureja, S. Gupta, A.K. Madan, Predicting anti-HIV-1 activity of 6-arylbenzonitriles: computational approach using super augmented eccentric connectivity topochemical indices, J. Mol. Graph. Model, 26 (2008) 1020-1029.

\bibitem{h11} B. Eskender, E. Vumar, Eccentric connectivity index and eccentric distance sum of some graph operations, Trans. Comb., 2(1)(2013) 103-111 .

\bibitem{h9} M. Eliasi, B. Taeri, Four new sums of graphs and their Wiener indices, Discrete Appl.  Math., 157(2009) 794-803.

\bibitem{t12} K. Fathalikhani, H. Faramarzi and H. Yousefi-Azari, Total eccentricity of some graph operations, Electronic Notes in Discrete. Math., 45(2014) 125-131.


\bibitem{i7} S. Fajtlowicz, On conjectures of graffiti II, Congr. Numer., 60 (1987) 189-197.

\bibitem{lso10} S. Gupta, M. Singh and A.K. Madan, Eccentric distance sum: a novel graph invariant for predicting biological and physical properties, J. Math. Anal. Appl., 275 (2002) 386-401.


\bibitem{lso12} H.B. Hua, K.X. Xu and W.N. Shu, A short and unified proof of Yu et al.�s two results on the eccentric distance sum, J. Math. Anal. Appl., 382 (2011) 364-366. 

\bibitem{lso13} H.B. Hua, S.G. Zhang, K.X. Xu, Further results on the eccentric distance sum, Discrete Appl. Math., 160 (2012) 170-180.

\bibitem{lso14} A. Ili\'c, I. Gutman, Eccentric connectivity index of chemical trees, MATCH Commun. Math. Comput. Chem., 65 (3) (2011) 731-744.

\bibitem{lso15} A. Ili\'c, G.H. Yu and L.H. Feng, On the eccentric distance sum of graphs, J. Math. Anal. Appl., 381 (2011) 590-600.

\bibitem{lso18} S.C. Li, M. Zhang, G.H. Yu and L.H. Feng, Extremal values on the eccentric distance sum of trees, J. Math. Anal. Appl., 390 (2012) 99-112.

\bibitem{h14} S. Li, G. Wang, Vertex PI indices of four sums of graphs, Discrete Appl. Math., 159(2011) 1601-1607.

\bibitem{fso1} Mehar Ali Malik, Two degree-distance based topological descriptors of some product graphs, Discrete Appl. Math.,(2017).

\bibitem{h16} M. Metsidik, W. Zhang and F. Duan, Hyper and reverse Wiener indices of F-sums of graphs, Discrete Appl. Math., 158(2010) 1433-1440.

\bibitem{i2} Nilanjan De, Sk. Md. Abu Nayeem and Anita Pal, Total Eccentricity index of the Generalized Hierarchical product of graphs, Int. J. Appl. Comput. Math., 1(2015) 503-511.


\bibitem{f} K. Pattabiraman, Inverse sum indeg index of graphs, AKCE Int. J. Graphs and Combin., 15(2018)155-167.


\bibitem{lso21} S. Sardana, A.K. Madan, Application of graph theory: relationship of molecular connectivity index, Wiener�s index and eccentric connectivity index with diuretic activity, MATCH Commun. Math. Comput. Chem., 43 (2001) 85-98.

\bibitem{lso22} S. Sardana, A.K. Madan, Predicting anti-HIV activity of TIBO derivatives: a computational approach using a novel topological descriptor, J. Mol. Model., 8 (2002) 258-265.

\bibitem{t16}Y. Tang, B. Zhou, On average eccentricity, MATCH Commun. Math. Comput. Chem., 67 (2012) 405-423.

\bibitem{lso27} G.H. Yu, L.H. Feng and A. Ili\'c, On the eccentric distance sum of trees and unicyclic graphs, J. Math. Anal. Appl., 375 (2011) 99-107.

\bibitem{lso26} H. Wiener, Structural determination of paraffin boiling points, J. Am. Chem. Soc., 69 (1947) 17-20.  
\end{thebibliography}

\title {Inverse Connective Eccentricity Index and its Applications}
\author{ 
		 K. Pattabiraman\footnote{E\_mail: pramank@gmail.com} and T. Suganya\footnote{E\_mail: suganyatpr@gmail.com}\\
		 Department of Mathematics\\
		 Annamalai University\\
		 Annamalainagar 608 002\\
		 India		
		 }		 	 
\date{}
\maketitle
\begin{center}
{\bf Abstract}
\end{center} 
The inverse connective eccentricity index of a connected graph $G$ is defined as $\xi^{-1}_{ce}(G)=\sum\limits_{u\in V(G)}\frac{\epsilon_G(u)}{d_G(u)},$ where $\epsilon_G(u)$ and $d_G(u)$ are the eccentricity and degree of a vertex $u$ in $G,$ respectively. In this paper, we obtain an upper bounds for inverse connective  eccentricity indices for various classes of graphs such as  generalized hierarchical product graph and $F$-sum  of graphs.\\
\NI{\bf MSC:} 05C12, 05C76\\
\NI{\bf Keywords:} Eccentricity, Topological index, Product graph.
\section{Introduction}

All graphs considered in this paper are simple and connected. Let $N_G(v)$ be the set of all neighbors of a vertex $v$ in a graph $G.$ The degree $d_G(v)$ of a vertex $v$ in $G$ is the cardinality of the set $N_G(v).$ A vertex with degree one is called a pendent vertex. The eccentricity of a vertex $u,$ denoted by $\epsilon_G(u),$ is the largest distance from $u$ to any other vertex $v$ of $G.$ 

A topological index $Top(G)$ of $G$ is a real number with the property that for every graph $H$ isomorphic to $G,$ $Top(H) =Top(G).$ In organic Chemistry, topological indices have been found to be useful in chemical documentation, isomer discrimination, structure-property relationships, structure-activity relationships and pharmaceutical drug design. Wiener index is the first  distance-based topological index which was introduced by Wiener \cite{lso26} in 1947. Wiener used his index for the calculation of boiling points of alkanes. The Wiener index of a graph $G$ is denoted by $W(G)$ and defined as the sum of distances between all pairs of vertices in graph $G,$ that is, $W(G)=\sum\limits_{\left\{u,v\right\}\subseteq V(G)}d_G(u,v).$

In recent years, Some indices have been derived related to eccentricity such as eccentric connectivity index,eccentric distance sum, adjacent distance sum, total eccentricity index. The eccentric connectivity index was successfully used for mathematical models of biological activities of diverse nature \cite{lso1,lso8,lso21}. It has been shown to give a high degree of predictability of pharmaceutical properties, and provide leads for the development of safe and potent anti-HIV compounds \cite{lso22}. Eccentric connectivity index is also proposed as a measure of branching in alkanes \cite{lso14}.

Gupta et al. \cite{lso10} introduced a novel topological descriptor which is called eccentric-distance sum index. This index offers a vast potential for structure�activity/property relationships; it also displays high discriminating power with respect to both biological activity and physical properties \cite{lso10}. From \cite{lso10}, we also know that some structure�activity and quantitative structure�property studies using eccentric-distance sum were better than the corresponding values obtained using the Wiener index. More recently, the mathematical properties of eccentric-distance sum have been investigated. Yu et al. \cite{lso27} computed the EDS of trees and unicyclic graphs. Hua et al. \cite{lso12} obtained the sharp lower bound on EDS of $n$-vertex cacti. Hua et al. \cite{lso13} studied the graphs with graph parameters having the minimum EDS. Ili\'c et al. \cite{lso15} studied the various lower and upper bounds for the  EDS in terms of the other graph invariant. Li et al. \cite{lso18} determined the trees with the third and fourth minimal EDS among the $n$-vertex trees.

Motivated  above indices Malik \cite{fso1} proposed another  topological descriptors related to eccentricity. The inverse connective eccentricity index of  G is denoted by $\xi_{ce}^{-1}(G),$ is defined as $\xi_{ce}^{-1}(G)=\sum\limits_{u\in V(G)}\frac{\epsilon_G(u)}{d_G(u)}.$ Inverse degree index of $G$ is defined as $ID(G)=\sum\limits_{u\in V(G)}\frac{1}{d_G(u)}.$ For notions not defined here we refer the reader to \cite{i1}. The inverse degree first attracted attention through numerous conjectures generated by the computer programme Graffiti\cite{i7}.

A number of topological indices based on vertex eccentricity are already subject to various studies.
The total eccentricity index of G is defined as $\zeta(G)=\sum\limits_{u\in V(G)}\epsilon_G(u).$
Similar to this index, Dankelmann et. al. \cite{t3} and Tang et al. \cite{t16} studied average eccentricity of graphs. Fathalikhani et al. in \cite{t12}, studied total eccentricity of some graph operations. In this paper, we obtain an upper bounds for inverse connective eccentricity indices for various classes of graphs such as  generalized hierarchical product graph and $F$-sum graphs.

\section{Generalized hierarchical product of graphs }

 A graph $G$ with a specified vertex subset $U\subseteq V(G)$ is denoted by $G(U).$ Barriere
et al. \cite{h3,h4} defined a new product of graphs, namely, the generalized hierarchical
product, as follows: Let $G$ and $H$ be two graphs with a nonempty vertex subset
$U \subseteq V(G).$ Then the {\it generalized hierarchical product}, denoted by $G(U)\sqcap H,$ is the
graph with vertex set $V(G)\times V(H)$ and two vertices $(g, h)$ and $(g^{'}, h^{'})$ are adjacent
if and only if $g = g^{'} \in U$ and $hh^{'} \in E(H)$ or, $gg^{'} \in E(G)$ and $h = h^{'}$.
{\it The Cartesian product}, $G\Box H$ of graphs $G$ and $H$ has the vertex set $V(G\Box H) =
V(G)\times V(H)$ and $(u, x)(v,y)$ is an edge of$G\Box H$ if $u = v$ and $xy \in E(H)$ or, $uv \in E(G)$
and $x=y$.

To each vertex $u \in V(G),$ there is an isomorphic copy of $H$ in $G \Box H$ and to each
vertex $v \in V(H),$ there is an isomorphic copy of $G$ in $G \Box H.$ But in the generalized hierarchical product, to each vertex $u \in U,$ there is an isomorphic copy of $H$ and to
each vertex $v \in V(H),$ there is an isomorphic copy of $G.$ In particular, if $U = V(G).$
then $G \Box H = G(U) \sqcap H.$ The generalized hierarchical product and Cartesian product of $P_5$ and $P_4$ are shown in Figure 1. 
\begin{center}
% Generated with LaTeXDraw 2.0.8
% Tue Jun 05 17:10:27 IST 2018
% \usepackage[usenames,dvipsnames]{pstricks}
% \usepackage{epsfig}
% \usepackage{pst-grad} % For gradients
% \usepackage{pst-plot} % For axes
\scalebox{1} % Change this value to rescale the drawing.
{
\begin{pspicture}(0,-3.3789062)(13.300938,3.3589063)
\psline[linewidth=0.04cm](1.8209375,2.6589062)(1.7809376,-1.5610938)
\psline[linewidth=0.04cm](3.0809374,2.6589062)(3.1009376,-1.5210937)
\psline[linewidth=0.04cm](4.2809377,2.6389062)(4.3009377,-1.5410937)
\psline[linewidth=0.04cm](5.5609374,2.6789062)(5.5609374,-1.5410937)
\psdots[dotsize=0.12](1.8209375,2.6189063)
\psdots[dotsize=0.12](3.0809374,2.6589062)
\psdots[dotsize=0.12](4.2809377,2.6189063)
\psdots[dotsize=0.12](5.5809374,2.6389062)
\psdots[dotsize=0.12](1.7609375,-1.5410937)
\psdots[dotsize=0.12](3.1009376,-1.5210937)
\psdots[dotsize=0.12](4.3009377,-1.5210937)
\psdots[dotsize=0.12](5.5609374,-1.5410937)
\psdots[dotsize=0.12](1.8009375,1.6589062)
\psdots[dotsize=0.12](1.8409375,0.63890624)
\psdots[dotsize=0.12](1.8009375,-0.40109375)
\psdots[dotsize=0.12](3.0809374,-0.34109375)
\psdots[dotsize=0.12](3.1009376,0.6589062)
\psdots[dotsize=0.12](3.0809374,1.6389062)
\psdots[dotsize=0.12](4.2609377,1.6589062)
\psdots[dotsize=0.12](4.3009377,0.63890624)
\psdots[dotsize=0.12](4.2809377,-0.36109376)
\psdots[dotsize=0.12](5.5809374,-0.32109374)
\psdots[dotsize=0.12](5.5209374,0.63890624)
\psdots[dotsize=0.12](5.5409374,1.6589062)
\psline[linewidth=0.04cm](1.9009376,0.6589062)(5.4409375,0.6589062)
\psline[linewidth=0.04cm](1.8609375,-0.38109374)(5.5209374,-0.36109376)
\psline[linewidth=0.04cm](1.8409375,1.6789062)(5.4609375,1.6989063)
\usefont{T1}{ptm}{m}{n}
\rput(2.8723438,-2.2510939){$ P_5(U)\Pi P_4,$ where$ U=\{2,3,4\}.$}
\psline[linewidth=0.04cm](9.600938,2.7389061)(9.560938,-1.4810938)
\psline[linewidth=0.04cm](10.800938,2.7389061)(10.820937,-1.4410938)
\psline[linewidth=0.04cm](12.000937,2.7189062)(12.020938,-1.4610938)
\psline[linewidth=0.04cm](13.200937,2.7589064)(13.200937,-1.4610938)
\psdots[dotsize=0.12](9.600938,2.6989062)
\psdots[dotsize=0.12](10.800938,2.7389061)
\psdots[dotsize=0.12](12.000937,2.6989062)
\psdots[dotsize=0.12](13.220938,2.7189062)
\psdots[dotsize=0.12](9.540937,-1.4610938)
\psdots[dotsize=0.12](10.820937,-1.4410938)
\psdots[dotsize=0.12](12.020938,-1.4410938)
\psdots[dotsize=0.12](13.200937,-1.4610938)
\psdots[dotsize=0.12](9.580937,1.7389063)
\psdots[dotsize=0.12](9.620937,0.7189062)
\psdots[dotsize=0.12](9.580937,-0.32109374)
\psdots[dotsize=0.12](10.800938,-0.26109374)
\psdots[dotsize=0.12](10.820937,0.73890626)
\psdots[dotsize=0.12](10.800938,1.7189063)
\psdots[dotsize=0.12](11.980938,1.7389063)
\psdots[dotsize=0.12](12.020938,0.7189062)
\psdots[dotsize=0.12](12.000937,-0.28109375)
\psdots[dotsize=0.12](13.220938,-0.24109375)
\psdots[dotsize=0.12](13.160937,0.7189062)
\psdots[dotsize=0.12](13.180938,1.7389063)
\psline[linewidth=0.04cm](9.620937,0.73890626)(13.160937,0.73890626)
\psline[linewidth=0.04cm](9.580937,-0.30109376)(13.240937,-0.28109375)
\psline[linewidth=0.04cm](9.560938,1.7589062)(13.180938,1.7789062)
\usefont{T1}{ptm}{m}{n}
\rput(11.062344,-2.0910938){ $P_5 \Box P_4$}
\psline[linewidth=0.04cm](9.600938,2.6989062)(13.240937,2.7389061)
\psline[linewidth=0.04cm](9.540937,-1.4610938)(13.200937,-1.4610938)
\psline[linewidth=0.04cm](1.8809375,3.2989063)(5.5209374,3.2789063)
\psline[linewidth=0.04cm](0.8809375,2.6589062)(0.8809375,-1.5610938)
\psdots[dotsize=0.12](1.8809375,3.2789063)
\psdots[dotsize=0.12](3.0209374,3.2389061)
\psdots[dotsize=0.12](4.3009377,3.2189062)
\psdots[dotsize=0.12](5.5209374,3.2589064)
\psdots[dotsize=0.12](0.8809375,2.5989063)
\psdots[dotsize=0.12](0.8809375,1.6789062)
\psdots[dotsize=0.12](0.8809375,0.67890626)
\psdots[dotsize=0.12](0.8809375,-0.40109375)
\psdots[dotsize=0.12](0.9209375,-1.5610938)
\usefont{T1}{ptm}{m}{n}
\rput(0.67234373,2.6489062){$1$}
\usefont{T1}{ptm}{m}{n}
\rput(0.5523437,1.7489063){$2$}
\usefont{T1}{ptm}{m}{n}
\rput(0.53234375,0.70890623){$3$}
\usefont{T1}{ptm}{m}{n}
\rput(0.51234376,-0.45109376){$4$}
\usefont{T1}{ptm}{m}{n}
\rput(0.57234377,-1.5910938){$5$}
\usefont{T1}{ptm}{m}{n}
\rput(7.400156,-3.1510937){Figure 1.}
\end{pspicture} 
}


\end{center}

A path between the vertices $u, v \in V(G)$ through $U �\subseteq V(G)$ is a $uv$-path in $G$ containing
some vertex $x\in U$ (vertex $x$ could be the vertex $u$ or $v$). The distance between $u$ and $v$ through
$U$ is the length of the shortest path between $u$ and $v$ through $U$ and is denoted by $d_{G(U)}(u, v).$
If $u\in U$ then we have,
$\epsilon _{G(U)}(u) = max_{u�\in V(G)}d_{G(U)}(u,v)$ and $\xi^{-1}_{ce}(G(U))=\sum\limits_{u_i\in V(G)}\frac{\epsilon_{G(U)}(u_i)}{d_{G(U)}(u_i)}.$\\
From the structure of a graph $G(U)\sqcap H,$ we have the following Lemma.
\begin{lem}\label{l0}Let $G$ and $H$ be two connected graphs and $U \subseteq V(G).$ Then $\epsilon_{G(U)\sqcap H}((u,v))=\epsilon_{G(U)}(u)+\epsilon_{H}(v).$\hfill\eb
\end{lem}
\begin{lem}\cite{f}\label{l1}
Let $a$ and $b$ be real number. Then $\frac{1}{a+b}\leq \frac{1}{4}\Big(\frac{1}{a}+\frac{1}{b}\Big)$ with equality if and only if $a=b.$
\end{lem} 
\begin{thm}\label{th1}
Let $G$ and $H$ be two connected graphs and $U \subseteq V(G).$ Then
$\xi^{-1}_{ce}(G(U)\sqcap H) \leq \left|V(H)\right|\xi^{-1}_{ce}(G(U))+\frac{ID(H)}{4}\zeta(G(U))+ID(G(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).$
\end{thm}
\NI{\bf Proof:} By the definition of inverse connective eccentricity index
\begin{eqnarray}
\xi^{-1}_{ce}(G(U)\sqcap H)&=&\sum\limits_{i=1}^{n}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)\sqcap H}((u_i,v_j))}{d_{G(U)\sqcap H}((u_i,v_j))}\nonumber\\
&=&\Big(\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}+\sum\limits_{u_i\in V(G)|U}\sum\limits_{j=1}^{m}\Big)\frac{\epsilon_{G(U)\sqcap H}((u_i,v_j))}{d_{G(U)\sqcap H}((u_i,v_j))}\nonumber\\
&=&\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)}(u_i)+\epsilon_H(v_j)}{d_G(u_i)+d_H(v_j)}+\sum\limits_{u_i\in V(G)|U}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)}(u_i)+\epsilon_H(v_j)}{d_G(u_i)},\text{~~ by Lemma \ref{l0}}\nonumber
\end{eqnarray}
By Lemma \ref{l1}, we have $\frac{1}{d_G(u_i)+d_H(v_j)}\leq\frac{1}{4}\Big(\frac{1}{d_G(u_i)}+\frac{1}{d_H(v_j)}\Big)$
with equality if and only if $d_G(u_i)=d_H(v_j).$ Hence
\begin{eqnarray}
\xi^{-1}_{ce}(G(U)\sqcap H)&\leq&\frac{1}{4}\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}(\epsilon_{G(U)}(u_i)+\epsilon_H(v_j))\Big(\frac{1}{d_G(u_i)}+\frac{1}{d_H(v_i)}\Big)\nonumber\\&+&\sum\limits_{u_i\in V(G)}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)}(u_i)+\epsilon_H(v_j)}{d_G(u_i)}\nonumber\\
&=&\frac{1}{4}\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)}(u_i)}{d_G(u_i)}+\frac{1}{4}\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)}(u_i)}{d_H(v_j)}+\frac{1}{4}\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}\frac{\epsilon_{H}(v_j)}{d_G(u_i)}\nonumber\\
&+&\frac{1}{4}\sum\limits_{u_i\in U}\sum\limits_{j=1}^{m}\frac{\epsilon_{H}(v_j)}{d_H(v_j)}+\sum\limits_{u_i\in V(G)|U}\sum\limits_{j=1}^{m}\frac{\epsilon_{G(U)}(u_i)}{d_G(u_i)}+\sum\limits_{u_i\in V(G)|U}\sum\limits_{j=1}^{m}\frac{\epsilon_{H}(v_j)}{d_G(u_i)}\nonumber
\end{eqnarray}
From the definition of inverse degree, we obtain
\begin{eqnarray}
\xi^{-1}_{ce}(G(U)\sqcap H)&\leq&\frac{1}{4}\left|V(H)\right|\sum\limits_{u_i\in U}\frac{\epsilon_{G(U)}(u_i)}{d_G(u_i)}+\frac{1}{4}ID(H)\sum\limits_{u_i\in U}\epsilon_{G(U)}(u_i)\nonumber\\&+&\frac{1}{4}\sum\limits_{u_i\in U}\frac{1}{d_G(u_i)}\sum\limits_{j=1}^{m}\epsilon_{H}(v_j)+\frac{1}{4}\left|U\right|\sum\limits_{j=1}^{m}\frac{\epsilon_{H}(v_j)}{d_H(v_j)}\nonumber\\
&+&\left|V(H)\right|\sum\limits_{u_i\in V(G)|U}\frac{\epsilon_{G(U)}(u_i)}{d_G(u_i)}+\sum\limits_{u_i\in V(G)|U}\frac{1}{d_G(u_i)}\sum\limits_{j=1}^{m}\epsilon_{H}(v_j)\nonumber\\
&=&\frac{\left|V(H)\right|}{4}\sum\limits_{u_i\in U}\frac{\epsilon_{G(U)}(u_i)}{d_G(u_i)}+\frac{ID(H)}{4}\zeta(G(U))+\frac{1}{4}\sum\limits_{u_i\in U}\frac{1}{d_G(u_i)}\zeta(H)\nonumber\\
&+&\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H)+\left|V(H)\right|\sum\limits_{u_i\in V(G)|U}\frac{\epsilon_{G(U)}(u_i)}{d_G(u_i)}+\zeta(H)\sum\limits_{u_i\in V(G)|U}\frac{1}{d_G(u_i)}\nonumber
\end{eqnarray}
By the definition of inverse connective eccentricity index, we have
\begin{eqnarray}
\xi^{-1}_{ce}(G(U)\sqcap H)&\leq&\left|V(H)\right|\xi^{-1}_{ce}(G(U))+\frac{ID(H)}{4}\zeta(G(U))+ID(G(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).\nonumber\hfill\eb
\end{eqnarray}
By setting $U=V(G)$ in  Theorem \ref{th1} , we obtain the upper bound for inverse connective eccentricity index of Cartesian product of two graphs $G$ and $H.$
\begin{cor}\label{c1} Let $G$ and $H$ be a connected graph. Then
$\xi^{-1}_{ce}(G\Box H)\leq \left|V(H)\right|\xi^{-1}_{ce}(G)+\frac{ID(H)}{4}\zeta(G)+ID(G)\zeta(H)+\frac{\left|V(G)\right|}{4}\xi^{-1}_{ce}(H).$\hfill\eb
\end{cor}

\begin{exam}
The $TC_4(m,n)$ nanotorns is a graph which is isomorphic to  a Cartesian product of two cycles. Hence by Corollary \ref{c1}, we obtain
\begin{eqnarray}
\xi^{-1}_{ce}(TC_4(m,n))\leq\begin{cases}\frac{5mn(2m+n-3)}{16}\text{~~ if $n,m$ are odd,}&\,\\\nonumber
  \frac{mn(5m+6n-6)}{16} \text{~~if $n$ is odd $m$ is even,}&\,\\  
 \frac{5mn(m+n-1)}{16}   \text{~~if $n$ is even $m$ is odd,}&\,\\ \frac{5mn(m+n)}{16} \text{~~if $n,m$ are even.}&\,\\
 \end{cases}
\end{eqnarray}
\end{exam}

\section{F-sum of graphs}
Let $G$ be a connected graph. Then

$(i)$The {\it subdivision graph} of a graph $G,$ denoted by $S(G),$ is obtained from G by replacing
each edge of G by a path of length two.

$(ii)$ The {\it triangle parallel graph} of a graph $G$ is denoted by
$R(G)$ and is obtained from $G$ by replacing each edge of $G$ by a triangle.

$(iii)$ The {\it line superposition graph} $Q(G)$ of a graph $G$ is obtained from $G$ by inserting a new vertex into each edge of
$G$ and then joining with edges each pair of new vertices on adjacent edges of $G.$

$(iv)$ The {\it total
graph} $T (G)$ of a graph G has its vertices as the edges and vertices of $G$ and adjacency in
$T (G)$ is defined by the adjacency or incidence of the corresponding elements of $G,$ see Figure 2.

\begin{center}
% Generated with LaTeXDraw 2.0.8
% Tue Jun 19 12:03:32 IST 2018
% \usepackage[usenames,dvipsnames]{pstricks}
% \usepackage{epsfig}
% \usepackage{pst-grad} % For gradients
% \usepackage{pst-plot} % For axes
\scalebox{1} % Change this value to rescale the drawing.
{
\begin{pspicture}(0,-4.688906)(5.9,4.668906)
\pstriangle[linewidth=0.04,dimen=outer](2.67,3.1489062)(1.86,1.46)
\pstriangle[linewidth=0.04,dimen=outer](1.09,0.56890625)(1.86,1.78)
\pstriangle[linewidth=0.04,dimen=outer](4.4,0.60890627)(1.8,1.74)
\pstriangle[linewidth=0.04,dimen=outer](1.06,-2.8510938)(1.96,2.06)
\pstriangle[linewidth=0.04,dimen=outer](4.5,-2.7510939)(2.2,2.08)
\psdots[dotsize=0.16](2.66,4.5689063)
\psdots[dotsize=0.16](1.76,3.1689062)
\psdots[dotsize=0.16](3.56,3.1489062)
\psdots[dotsize=0.16](1.08,2.2689064)
\psdots[dotsize=0.16](0.2,0.56890625)
\psdots[dotsize=0.16](1.96,0.5889062)
\psdots[dotsize=0.16](4.42,2.3289063)
\psdots[dotsize=0.16](3.52,0.60890627)
\psdots[dotsize=0.16](5.28,0.60890627)
\psdots[dotsize=0.16](4.48,-0.7110937)
\psdots[dotsize=0.16](3.42,-2.7310936)
\psdots[dotsize=0.16](5.54,-2.7510939)
\psdots[dotsize=0.16](0.08,-2.7910938)
\psdots[dotsize=0.16](1.08,-0.81109375)
\psdots[dotsize=0.16](2.06,-2.7910938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.6,1.4489063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.52,1.4689063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.1,0.5889062)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](3.34,1.7889062)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](5.5,1.7689062)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.48,0.10890625)
\psline[linewidth=0.04cm](3.38,1.8289063)(4.4,2.3689063)
\psline[linewidth=0.04cm](3.34,1.7289063)(3.5,0.5889062)
\psline[linewidth=0.04cm](3.5,0.5889062)(4.4,0.08890625)
\psline[linewidth=0.04cm](4.54,0.10890625)(5.26,0.5889062)
\psline[linewidth=0.04cm](4.44,2.3489063)(5.4,1.8489063)
\psline[linewidth=0.04cm](5.5,1.7289063)(5.28,0.62890625)
\psline[linewidth=0.04cm](0.56,-1.8110938)(1.52,-1.8310938)
\psline[linewidth=0.04cm](1.52,-1.8310938)(1.14,-2.7910938)
\psline[linewidth=0.04cm](0.6,-1.8110938)(1.04,-2.7910938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.56,-1.7910937)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.52,-1.8310938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.08,-2.7910938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](3.16,-1.4310937)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](5.8,-1.4110937)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.4,-3.4310937)
\psline[linewidth=0.04cm](3.24,-1.4310937)(5.72,-1.4310937)
\psline[linewidth=0.04cm](5.72,-1.4310937)(4.44,-3.3710938)
\psline[linewidth=0.04cm](3.22,-1.4510938)(4.4,-3.3510938)
\psline[linewidth=0.04cm](3.2,-1.3710938)(4.48,-0.67109376)
\psline[linewidth=0.04cm](3.16,-1.4710938)(3.36,-2.6510937)
\psline[linewidth=0.04cm](3.4,-2.7510939)(4.34,-3.4310937)
\psline[linewidth=0.04cm](4.44,-3.4710937)(5.58,-2.7910938)
\psline[linewidth=0.04cm](5.56,-2.6710937)(5.84,-1.4110937)
\psline[linewidth=0.04cm](4.52,-0.63109374)(5.8,-1.3310938)
\usefont{T1}{ptm}{m}{n}
\rput(2.6314063,2.6789062){$G$}
\usefont{T1}{ptm}{m}{n}
\rput(1.1114062,-0.08109375){$S(G)$}
\usefont{T1}{ptm}{m}{n}
\rput(4.4514065,-0.18109375){$R(G)$}
\usefont{T1}{ptm}{m}{n}
\rput(1.0814062,-3.6210938){$Q(G)$}
\usefont{T1}{ptm}{m}{n}
\rput(4.4714065,-3.8010938){$T(G)$}
\usefont{T1}{ptm}{m}{n}
\rput(2.6392188,-4.461094){Figure 2.}
\end{pspicture} 
}


\end{center}

 Let $F$ be one of the subdivision operations $S, R, Q$ or $T.$ For two connected
graphs $G$ and $H,$ the $F-$sum, denoted by $G+_F H,$ is the graph with vertex set $(V (G)\cup
E(G))\times V(H),$ and any two vertices $(u, v)$ and $(u^{'},v^{'})$ of $G+_F H$ are adjacent if and only $\Big[u = u^{'}\in V(G) \text{~~and~~} (v, v^{'})\in E(H)\Big]$ or $\Big[ v= v^{'}\in V(H) \text{~~and~~} (u, u^{'})\in E(F(G))\Big].$ The graph $P_3+_FP_2$ are shown in Figure 3. 
\begin{center}
% Generated with LaTeXDraw 2.0.8
% Tue Jun 19 12:27:26 IST 2018
% \usepackage[usenames,dvipsnames]{pstricks}
% \usepackage{epsfig}
% \usepackage{pst-grad} % For gradients
% \usepackage{pst-plot} % For axes
\scalebox{1} % Change this value to rescale the drawing.
{
\begin{pspicture}(0,-4.578906)(8.26,4.558906)
\psline[linewidth=0.04cm](0.1,3.8989062)(0.1,2.5189064)
\psline[linewidth=0.04cm](1.48,3.8589063)(1.5,2.5589063)
\psline[linewidth=0.04cm](2.88,3.8589063)(2.88,2.5189064)
\psdots[dotsize=0.16](0.1,2.5189064)
\psdots[dotsize=0.16](1.48,2.5189064)
\psdots[dotsize=0.16](2.9,2.5189064)
\psdots[dotsize=0.16](2.88,3.8789062)
\psdots[dotsize=0.16](1.5,3.8589063)
\psdots[dotsize=0.16](0.1,3.8989062)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.78,4.458906)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.28,4.438906)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.78,1.8589063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.26,1.8589063)
\psline[linewidth=0.04cm](0.14,3.9589062)(0.74,4.438906)
\psline[linewidth=0.04cm](0.82,4.438906)(1.44,3.9189062)
\psline[linewidth=0.04cm](0.12,2.4989061)(0.68,1.8989062)
\psline[linewidth=0.04cm](0.84,1.8789062)(1.46,2.4789062)
\psline[linewidth=0.04cm](1.52,3.8589063)(2.2,4.418906)
\psline[linewidth=0.04cm](2.32,4.418906)(2.86,3.9189062)
\psline[linewidth=0.04cm](1.5,2.5389063)(2.16,1.8989062)
\psline[linewidth=0.04cm](2.32,1.8589063)(2.84,2.4589062)
\usefont{T1}{ptm}{m}{n}
\rput(1.5714062,1.3489063){$P_3+_SP_2$}
\psline[linewidth=0.04cm](0.08,-0.14109375)(0.1,-1.7210938)
\psline[linewidth=0.04cm](1.48,-0.12109375)(1.48,-1.7010938)
\psline[linewidth=0.04cm](2.86,-0.12109375)(2.9,-1.6410937)
\psdots[dotsize=0.16](0.08,-0.14109375)
\psdots[dotsize=0.16](0.1,-1.7210938)
\psdots[dotsize=0.16](1.5,-1.7010938)
\psdots[dotsize=0.16](1.48,-0.10109375)
\psdots[dotsize=0.16](2.84,-0.12109375)
\psdots[dotsize=0.16](2.9,-1.7010938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.8,0.49890625)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.24,0.51890624)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.74,-2.3610938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.28,-2.3210938)
\psline[linewidth=0.04cm](0.12,-0.08109375)(0.74,0.47890624)
\psline[linewidth=0.04cm](0.84,0.45890626)(1.44,-0.06109375)
\psline[linewidth=0.04cm](0.1,-1.7410938)(0.66,-2.3010938)
\psline[linewidth=0.04cm](0.8,-2.3210938)(1.48,-1.7410938)
\psline[linewidth=0.04cm](1.52,-0.04109375)(2.2,0.47890624)
\psline[linewidth=0.04cm](2.28,0.49890625)(2.8,-0.08109375)
\psline[linewidth=0.04cm](1.52,-1.7010938)(2.22,-2.3410938)
\psline[linewidth=0.04cm](2.34,-2.3610938)(2.92,-1.7210938)
\psline[linewidth=0.04cm](0.82,0.49890625)(2.16,0.51890624)
\psline[linewidth=0.04cm](0.8,-2.3210938)(2.16,-2.3210938)
\usefont{T1}{ptm}{m}{n}
\rput(1.5814062,-2.9910936){$P_3+_QP_2$}
\psline[linewidth=0.04cm](2.16,-3.0410938)(2.16,-3.0210938)
\psline[linewidth=0.04cm](5.36,3.8589063)(5.36,2.4789062)
\psline[linewidth=0.04cm](6.74,3.8189063)(6.76,2.5189064)
\psline[linewidth=0.04cm](8.14,3.8189063)(8.14,2.4789062)
\psdots[dotsize=0.16](5.36,2.4789062)
\psdots[dotsize=0.16](6.74,2.4789062)
\psdots[dotsize=0.16](8.16,2.4789062)
\psdots[dotsize=0.16](8.14,3.8389063)
\psdots[dotsize=0.16](6.76,3.8189063)
\psdots[dotsize=0.16](5.36,3.8589063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.04,4.418906)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](7.54,4.398906)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.04,1.8189063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](7.52,1.8189063)
\psline[linewidth=0.04cm](5.4,3.9189062)(6.0,4.398906)
\psline[linewidth=0.04cm](6.08,4.398906)(6.7,3.8789062)
\psline[linewidth=0.04cm](5.38,2.4589062)(5.94,1.8589063)
\psline[linewidth=0.04cm](6.1,1.8389063)(6.72,2.4389062)
\psline[linewidth=0.04cm](6.78,3.8189063)(7.46,4.3789062)
\psline[linewidth=0.04cm](7.58,4.3789062)(8.12,3.8789062)
\psline[linewidth=0.04cm](6.76,2.4989061)(7.42,1.8589063)
\psline[linewidth=0.04cm](7.58,1.8189063)(8.1,2.4189062)
\usefont{T1}{ptm}{m}{n}
\rput(6.731406,1.3689063){$P_3+_RP_2$}
\psline[linewidth=0.04cm](5.38,3.8989062)(8.14,3.8189063)
\psline[linewidth=0.04cm](5.34,2.4989061)(8.16,2.4989061)
\psline[linewidth=0.04cm](5.32,-0.34109375)(5.34,-1.9210937)
\psline[linewidth=0.04cm](6.72,-0.32109374)(6.72,-1.9010937)
\psline[linewidth=0.04cm](8.1,-0.32109374)(8.14,-1.8410938)
\psdots[dotsize=0.16](5.32,-0.34109375)
\psdots[dotsize=0.16](5.34,-1.9210937)
\psdots[dotsize=0.16](6.74,-1.9010937)
\psdots[dotsize=0.16](6.72,-0.30109376)
\psdots[dotsize=0.16](8.08,-0.32109374)
\psdots[dotsize=0.16](8.14,-1.9010937)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.04,0.29890624)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](7.48,0.31890625)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](7.52,-2.5210938)
\psline[linewidth=0.04cm](5.36,-0.28109375)(5.98,0.27890626)
\psline[linewidth=0.04cm](6.08,0.25890625)(6.68,-0.26109374)
\psline[linewidth=0.04cm](5.34,-1.9410938)(5.9,-2.5010939)
\psline[linewidth=0.04cm](6.04,-2.5210938)(6.72,-1.9410938)
\psline[linewidth=0.04cm](6.76,-0.24109375)(7.44,0.27890626)
\psline[linewidth=0.04cm](7.52,0.29890624)(8.04,-0.28109375)
\psline[linewidth=0.04cm](6.76,-1.9010937)(7.46,-2.5410938)
\psline[linewidth=0.04cm](7.58,-2.5610938)(8.16,-1.9210937)
\psline[linewidth=0.04cm](6.06,0.29890624)(7.4,0.31890625)
\psline[linewidth=0.04cm](5.3,-0.32109374)(8.04,-0.30109376)
\psline[linewidth=0.04cm](5.34,-1.8610938)(8.1,-1.8810937)
\psline[linewidth=0.04cm](5.98,-2.5610938)(7.44,-2.5410938)
\usefont{T1}{ptm}{m}{n}
\rput(6.5914063,-3.0710938){$P_3+_TP_2$}
\usefont{T1}{ptm}{m}{n}
\rput(3.9807813,-4.351094){Figure 3. F-sum graphs}
\psdots[dotsize=0.12,fillstyle=solid,dotstyle=o](5.98,-2.5410938)
\end{pspicture} 
}


\end{center}

The concept of $F-$sum graph was first introduced by Eliasi and Taeri \cite {h9} and the Weiner indices
of the resulting graphs were studied therein. Li and Wang \cite{h14} derived explicit expression
of the PI indices of four sums of two graphs. The hyper and reverse Weiner indices of F sum
graphs were studied by Metsidik et al. \cite {h16}. Eskender and Vumar \cite {h11} determined the
eccentric connectivity index of $F-$sum graphs in terms of some invariants of the factors. In this sequence,
now find the upper bounds for inverse connective eccentricity index of $G+_FH,$ where$ F=S,R,Q$ and $T.$

\begin{lem}\label{l2}\cite{i2} Let $G$ and $H$ be  two connected graphs. Then\\ (i)~~$\zeta(S(G)(U))=2\zeta(G)+\zeta(L(G))+\left|E(G)\right|,$\\
(ii)~~$\zeta(R(G)(U))=\zeta(G)+\zeta(L(G))+\left|E(G)\right|,$\\
(iii)~~$\zeta(Q(G)(U))=\zeta(G)+\zeta(L(G))+\left|V(G)\right|+\left|E(G)\right|,$\\
(iv)~~$\zeta(T(G)(U))=\zeta(G)+\zeta(L(G))+\left|E(G)\right|.$\hfill\eb
\end{lem}
\begin{thm}\label{t1} Let $G$ and $H$ be two connected graphs. Then
$\xi^{-1}_{ce}(G+_SH) \leq 2\left|V(H)\right|\xi^{-1}_{ce}(G)+\frac{\zeta(L(G))}{4}\Big(\left|V(H)\right|+ID(H)\Big)+\frac{\left|E(G)\right|}{4}(2\left|V(H)\right|+ID(H))+\frac{ID(H)}{2}\zeta(G)+ID(S(G)(U))\zeta(H)+\frac{\left|V(G)\right|+\left|E(G)\right|}{4}\xi^{-1}_{ce}(H).$ \end{thm}
\NI{\bf Proof:} Let $U$ be a subset of $V(S(G))$. From the structure of $S(G),$ the number of vertices and edge of $S(G)$ are $\left|V(G)\right| + \left|E(G)\right|$ and $2\left|E(G)\right|,$ respectively. Further, for each vertex $v\in U, $ $\epsilon_{S(G)(U)}(v) = 2\epsilon_G(v)$ and for each vertex $v \in V(S(G))|U,$ $\epsilon_{S(G)(U)}(v) = 2\epsilon_{L(G)}(v) + 1,$ where $L(G)$ is a line graph of $G$.
Hence by the definition of inverse connective eccentricity index, we have \\  
\begin{eqnarray}
\xi^{-1}_{ce}(S(G)(U))&=&\sum\limits_{v\in V(S(G)(U))}\frac{\epsilon_{S(G)(U)}(v)}{d_{S(G)(U)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{S(G)(U)}(v)}{d_{S(G)(U)}(v)}+\sum\limits_{v\in V(S(G))|U}\frac{\epsilon_{S(G)(U)}(v)}{d_{S(G)(U)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{2\epsilon_{G}(v)}{d_G(v)}+\sum\limits_{v\in V(S(G))|U}\frac{\epsilon_{S(G)(U)}(v)}{2}\nonumber\\
&=&2\xi^{-1}_{ce}(G)+\frac{1}{2}\sum_{v\in V(L(G))}(2\epsilon_{L(G)}(v)+1)\nonumber\\
&=&2\xi^{-1}_{ce}(G)+\zeta(L(G))+\frac{\left|E(G)\right|}{2}.\nonumber
\end{eqnarray}
Combining this result with result of $\xi^{-1}(G(U)\sqcap H)$ and by use of Lemma \ref{l2}, we obtain\\
$\xi^{-1}_{ce}(G+_SH)\leq 2\left|V(H)\right|\xi^{-1}_{ce}(G)+\frac{\zeta(L(G))}{4}\Big(\left|V(H)\right|+ID(H)\Big)+\frac{\left|E(G)\right|}{4}(2\left|V(H)\right|+ID(H))+\frac{ID(H)}{2}\zeta(G)+ID(S(G)(U))\zeta(H)+\frac{\left|V(G)\right|+\left|E(G)\right|}{4}\xi^{-1}_{ce}(H).$\hfill\eb

\begin{exam}
 The Zig-Zag polyhex nanotube (Figure 4) is the graph $S(C_n)(U)\sqcap P_2,$ where $U=V(C_n)\subseteq V(S(C_n)).$ 
 Hence by Theorem \ref{t1}, we obtain the following
 \begin{eqnarray}
\xi^{-1}_{ce}(S(C_n(U)\sqcap P_2))\leq\begin{cases}\frac{8n^3+3n^2+7n}{4}\text{~~ if $n$  odd,}&\,\\\nonumber 
 \frac{11n^2+18n}{4}   \text{~~if $n$ is even.}&\,\\ 
 \end{cases}
\end{eqnarray} 
\end{exam}
 \begin{center}
% Generated with LaTeXDraw 2.0.8
% Mon Jul 09 13:06:28 IST 2018
% \usepackage[usenames,dvipsnames]{pstricks}
% \usepackage{epsfig}
% \usepackage{pst-grad} % For gradients
% \usepackage{pst-plot} % For axes
\scalebox{1} % Change this value to rescale the drawing.
{
\begin{pspicture}(0,-1.6389062)(8.635938,1.6189063)
\psellipse[linewidth=0.04,dimen=outer](3.6940625,0.7189062)(3.52,0.84)
\psellipse[linewidth=0.04,dimen=outer](3.7740624,-0.02109375)(3.52,0.84)
\psdots[dotsize=0.16](2.6140625,1.4789063)
\psdots[dotsize=0.16](2.5740626,0.7789062)
\psdots[dotsize=0.16](4.3940625,1.5189062)
\psdots[dotsize=0.16](4.3940625,0.81890625)
\psdots[dotsize=0.16](1.3340625,1.3389063)
\psdots[dotsize=0.16](1.3940625,0.57890624)
\psdots[dotsize=0.16](0.1940625,0.6589062)
\psdots[dotsize=0.16](0.2940625,-0.18109375)
\psdots[dotsize=0.16](1.5940624,-0.68109375)
\psdots[dotsize=0.16](1.5940624,0.03890625)
\psdots[dotsize=0.16](2.5940626,-0.08109375)
\psdots[dotsize=0.16](2.5940626,-0.7810938)
\psdots[dotsize=0.16](3.6140625,-0.14109375)
\psdots[dotsize=0.16](3.5940626,-0.8410938)
\psdots[dotsize=0.16](4.3940625,-0.08109375)
\psdots[dotsize=0.16](4.4140625,-0.8410938)
\psdots[dotsize=0.16](5.1940627,-0.04109375)
\psdots[dotsize=0.16](5.2140627,-0.82109374)
\psdots[dotsize=0.16](6.1540623,0.11890625)
\psdots[dotsize=0.16](6.1940627,-0.64109373)
\psdots[dotsize=0.16](6.8740625,0.37890625)
\psdots[dotsize=0.16](6.8940625,-0.38109374)
\psdots[dotsize=0.16](5.3940625,1.4589063)
\psdots[dotsize=0.16](5.3740625,0.75890625)
\psdots[dotsize=0.16](6.1940627,1.2789062)
\psdots[dotsize=0.16](6.1940627,0.57890624)
\psline[linewidth=0.04cm](2.6140625,1.4589063)(2.5940626,0.6589062)
\psline[linewidth=0.04cm](1.3140625,1.3789062)(1.3940625,0.55890626)
\psline[linewidth=0.04cm](0.1940625,0.63890624)(0.2540625,-0.20109375)
\psline[linewidth=0.04cm](1.6140625,0.03890625)(1.5940624,-0.6610938)
\psline[linewidth=0.04cm](2.5940626,-0.08109375)(2.5940626,-0.76109374)
\psline[linewidth=0.04cm](3.6140625,-0.14109375)(3.6140625,-0.8410938)
\psline[linewidth=0.04cm](4.3940625,-0.08109375)(4.4340625,-0.8410938)
\psline[linewidth=0.04cm](5.1940627,-0.02109375)(5.2340627,-0.8410938)
\psline[linewidth=0.04cm](6.1340623,0.13890626)(6.1740627,-0.62109375)
\psline[linewidth=0.04cm](6.8540626,0.33890626)(6.8940625,-0.34109375)
\psline[linewidth=0.04cm](6.1740627,1.2789062)(6.1940627,0.5989063)
\psline[linewidth=0.04cm](5.3940625,1.4389062)(5.3740625,0.75890625)
\psline[linewidth=0.04cm](4.3740625,1.5789063)(4.3740625,0.79890627)
\psdots[dotsize=0.16](0.8540625,0.23890625)
\psdots[dotsize=0.16](0.7740625,-0.46109375)
\psdots[dotsize=0.16](7.1740627,0.7789062)
\psdots[dotsize=0.16](7.2540627,0.01890625)
\psline[linewidth=0.04cm](0.8140625,0.23890625)(0.7940625,-0.50109375)
\psline[linewidth=0.04cm](7.1740627,0.7789062)(7.2540627,-0.00109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.9940625,1.4389062)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.0140624,0.67890626)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.6340625,1.0989063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.7340625,0.35890624)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.4940625,0.39890626)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](0.4740625,-0.30109376)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.2140625,0.13890626)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](1.1540625,-0.56109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.1140625,-0.04109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](2.0940626,-0.76109374)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](3.1540625,-0.12109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](3.1540625,-0.82109374)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.0540624,-0.12109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.0740623,-0.8410938)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.8140626,-0.06109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.8940625,-0.80109376)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](5.6540623,0.01890625)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](5.7340627,-0.68109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.5340624,0.19890624)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.5740623,-0.50109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](7.0540624,0.51890624)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](7.1340623,-0.20109375)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.7740626,1.0989063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](6.6340623,0.47890624)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](5.8540626,1.3589063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](5.7740626,0.67890626)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.8740625,1.4789063)
\psdots[dotsize=0.16,fillstyle=solid,dotstyle=o](4.9340625,0.7789062)
\psdots[dotsize=0.12](3.2140625,1.1389062)
\psdots[dotsize=0.12](3.5140624,1.1389062)
\psdots[dotsize=0.12](3.7540624,1.1389062)
\usefont{T1}{ptm}{m}{n}
\rput(4.2920313,-1.4110937){Figure 4. The Zig-Zag polyhex nanotube $TUHC_6[2n,2].$}
\end{pspicture} 
}


\end{center}

\begin{thm} Let $G$ and $H$ be two connected graphs. Then
$\xi^{-1}_{ce}(G+_RH) \leq \left|V(H)\right|\Big(\xi^{-1}_{ce}(G)+\frac{1}{2}(2\zeta(L(G))+{\left|E(G)\right|}\Big)+\frac{ID(H)}{4}\Big(\zeta(R(G))+\zeta(L(G))+\left|E(G)\right|\Big)+ID(R(G)(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).$ \end{thm}
\NI{\bf Proof:} Let $U$ be a subset of $V(R(G)).$ From the structure of $R(G),$  we get $\left|V(R(G))\right|=\left|V(G)\right| + \left|E(G)\right|$ and $\left|E(R(G))\right|=3\left|E(G)\right|.$  Further, for each vertex $v\in U,$ $\epsilon_{R(G)(U)}(v) = \epsilon_G(v)$ and for each vertex $v \in V(R(G))|U,$ $\epsilon_{R(G)(U)}(v) = \epsilon_{L(G)}(v) + 1,$ where $L(G)$ is a line graph of $G$.
Thus by the definition of inverse connective eccentricity index, we obtain 
\begin{eqnarray}
\xi^{-1}_{ce}(R(G)(U))&=&\sum\limits_{v\in V(R(G))}\frac{\epsilon_{R(G)(U)}(v)}{d_{R(G)(U)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{R(G)(U)}(v)}{d_{R(G)(U)}(v)}+\sum\limits_{v\in V(R(G))|U}\frac{\epsilon_{R(G)(U)}(v)}{d_{R(G)(U)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{G}(v)}{d_G(v)}+\sum\limits_{v\in V(R(G))|U}\frac{\epsilon_{R(G)(U)}(v)}{2}\nonumber\\
&=&\xi^{-1}_{ce}(G)+\frac{1}{2}\sum_{v\in V(L(G))}(2\epsilon_{L(G)}(v)+1)\nonumber\\
&=&\xi^{-1}_{ce}(G)+\frac{1}{2}(\zeta(L(G))+{\left|E(G)\right|}).\nonumber
\end{eqnarray}
Combining this result with result of $\xi^{-1}(G(U)\sqcap H)$ and by use of Lemma \ref{l2}, we obtain

$\xi^{-1}_{ce}(G+_RH)\leq \left|V(H)\right|\Big(\xi^{-1}_{ce}(G)+\frac{1}{2}(2\zeta(L(G))+{\left|E(G)\right|}\Big)+\frac{ID(H)}{4}\Big(\zeta(G)+\zeta(L(G))+\left|E(G)\right|\Big)+ID(R(G)(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).$\hfill\eb
\begin{thm} Let $G$ and $H$ be two connected graphs. Then
$\xi^{-1}_{ce}(G+_QH) \leq \left|V(H)\right|\Big(\xi^{-1}_{ce}(G)+ID(G)+\frac{1}{4}\Big(\xi^{-1}_{ce}(L(G))+ID(L(G))+\frac{1}{2}\Big(\zeta(L(G))+\left|E(G)\right|\Big)\Big)\Big)+\frac{ID(H)}{4}\Big(\zeta(G)+\left|V(G)\right|+\zeta(L(G))+\left|E(G)\right|\Big)+ID(Q(G)(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H)  .$ \end{thm}

\NI{\bf Proof:} Let $U$ be a subset of $V(Q(G)).$ From the structure of $Q(G),$ the number of vertices and edge of $Q(G)$ are $\left|V(G)\right| + \left|E(G)\right|$ and $2\left|E(G)\right|+\left|E(L(G))\right|,$ respectively. Further, for each vertex $v\in U,$ $\epsilon_{Q(G)(U)}(v) = \epsilon_G(v)+1$ and for each vertex $v \in V(Q(G))|U,$ $\epsilon_{Q(G)(U)}(v) =\epsilon_{L(G)}(v) + 1,$ where $L(G)$ is a line graph of $G.$
Hence by the definition of inverse connective eccentricity index, we have\\ 
\begin{eqnarray}
\xi^{-1}_{ce}(Q(G)(U))&=&\sum\limits_{v\in V(Q(G))}\frac{\epsilon_{Q(G)(U)}(v)}{d_{Q(G)(U)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{Q(G)(U)}(v)}{d_{Q(G)(U)}(v)}+\sum\limits_{v\in V(Q(G))|U}\frac{\epsilon_{Q(G)(U)}(v)}{d_{Q(G)(U)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{G}(v)+1}{d_G(v)}+\sum\limits_{v\in V(L(G))}\frac{\epsilon_{L(G)(v)}+1}{d_{L(G)}(v)+2}.\nonumber
\end{eqnarray}
By Lemma \ref{l1}, we have $\frac{1}{d_{L(G)(v)+2}}\leq \frac{1}{4}\Big(\frac{1}{d_{L(G)}(v)}+\frac{1}{2}\Big)$ with equality if and only if $d_{L(G)}(v)=2.$ Hence
\begin{eqnarray}
\xi^{-1}_{ce}(Q(G)(U))&\leq&\sum\limits_{v\in U}\frac{\epsilon_G(v)}{d_G(v)}+\sum\limits_{v\in U}\frac{1}{d_G}(v)+\frac{1}{4}\Big(\sum\limits_{v\in V(L(G))}\frac{\epsilon_{L(G)(v)}+1}{d_{L(G)}(v)}+\sum\limits_{v\in V(L(G))}\frac{\epsilon_{L(G)(v)}+1}{2}\Big)\nonumber\\
&\leq&\xi^{-1}_{ce}(G)+ID(G)+\frac{1}{4}\Big(\xi^{-1}_{ce}(L(G))+ID(L(G))+\frac{1}{2}\Big(\zeta(L(G))+\left|E(G)\right|\Big)\Big).\nonumber
\end{eqnarray}
Combining this result with result of $\xi^{-1}(G(U)\sqcap H)$ and by use of Lemma \ref{l2}, we obtain

$\xi^{-1}_{ce}(G+_QH)\leq \left|V(H)\right|\Big(\xi^{-1}_{ce}(G)+ID(G)+\frac{1}{4}\Big(\xi^{-1}_{ce}(L(G))+ID(L(G))+\frac{1}{2}\Big(\zeta(L(G))+\left|E(G)\right|\Big)\Big)\Big)+\frac{ID(H)}{4}\Big(\zeta(G)+\left|V(G)\right|+\zeta(L(G))+\left|E(G)\right|\Big)+ID(Q(G)(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).$\hfill\eb

\begin{thm} Let $G$ and $H$ be two connected graphs. Then
$\xi^{-1}_{ce}(G+_TH) \leq \left|V(H)\right|\Big(2\xi^{-1}_{ce}(G)+\frac{1}{4}\Big(\xi^{-1}_{ce}(L(G))+ID(L(G))+\frac{\zeta(L(G))}{2}+\frac{\left|E(G)\right|}{2}\Big)+\frac{ID(H)}{4}\Big(\zeta(G)+\zeta(L(G))+\left|E(G)\right|\Big)+ID(T(G)(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).$ \end{thm}

\NI{\bf Proof:} Let $U$ be a subset of $V(T(G)).$ From the structure of $T(G),$ the number of vertices and edge of $T(G)$ are $\left|V(G)\right| + \left|E(G)\right|$ and $3\left|E(G)\right|+\left|E(L(G))\right|,$ respectively. Further, for each vertex $v\in U,$ $\epsilon_{T(G)(U)}(v) = \epsilon_G(v)$ and for each vertex $v \in V(T(G))|U,$ $\epsilon_{T(G)(U)}(v) =\epsilon_{L(G)}(v) + 1,$ where $L(G)$ is a line graph of $G.$
Thus by the definition of inverse connective eccentricity index, we get 
\begin{eqnarray}
\xi^{-1}_{ce}(T(G)(U))&=&\sum\limits_{v\in V(T(G))}\frac{\epsilon_{T(G)(U)}(v)}{d_{T(G)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{T(G)(U)}(v)}{d_{T(G)}(v)}+\sum\limits_{v\in V(T(G))|U}\frac{\epsilon_{T(G)(U)}(v)}{d_{T(G)}(v)}\nonumber\\
&=&\sum\limits_{v\in U}\frac{\epsilon_{G}(v)}{d_G(v)}+\sum\limits_{v\in V(L(G))}\frac{\epsilon_{L(G)(v)}+1}{d_{L(G)}+2}.\nonumber
\end{eqnarray}

By Lemma \ref{l1}, we have $\frac{1}{d_{L(G)(v)+2}}\leq \frac{1}{4}\Big(\frac{1}{d_{L(G)}(v)}+\frac{1}{2}\Big)$ with equality if and only if $d_{L(G)}(v)=2.$ Therefore

\begin{eqnarray}
\xi^{-1}_{ce}(T(G)(U))&\leq&2\xi^{-1}_{ce}(G)+\frac{1}{4}\Big(\sum\limits_{v\in L(G)}\frac{\epsilon_{L(G)}(v)+1}{d_L(G)(v)}+\frac{\epsilon_L(G)+1}{2}\Big)\nonumber\\
&\leq&2\xi^{-1}_{ce}(G)+\frac{1}{4}\Big(\xi^{-1}_{ce}(L(G))+ID(L(G))+\frac{\zeta(L(G))}{2}+\frac{\left|E(G)\right|}{2}\Big).\nonumber
\end{eqnarray}
Combining this result with result of $\xi^{-1}(G(U)\sqcap H)$ and by the use of Lemma \ref{l2}, we obtain

$\xi^{-1}_{ce}(G+_TH)\leq \left|V(H)\right|\Big(2\xi^{-1}_{ce}(G)+\frac{1}{4}\Big(\xi^{-1}_{ce}(L(G))+ID(L(G))+\frac{\zeta(L(G))}{2}+\frac{\left|E(G)\right|}{2}\Big)+\frac{ID(H)}{4}\Big(\zeta(G)+\zeta(L(G))+\left|E(G)\right|\Big)+ID(T(G)(U))\zeta(H)+\frac{\left|U\right|}{4}\xi^{-1}_{ce}(H).$\hfill\eb



\end{document}